% \documentstyle[12pt]{report}
\documentclass[12pt]{article}
\usepackage{exscale}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{graphics}
%
\include{definitions}
\textheight24truecm
\begin{document}
This is a list of (important) known errors in the notes, not corrected in the
hand-outs, but corrected in the online version.

\begin{enumerate}
\item page 8: replaced $n=1-b/a$ with $n=10(1-b/a)$
\item page 52: winding problem, replace Period $P\propto 1/R$ with
Period $P\propto R$.
\end{enumerate}


\noindent {\bf Think you found an error? Let me know, tom.theuns@durham.ac.uk}

\end{document}



